UC Berkeley

UC Berkeley Previously Published Works

Title

Modeling environmentally mediated rotavirus transmission: The role of temperature and hydrologic factors

Permalink https://escholarship.org/uc/item/7q065245

Journal

Proceedings of the National Academy of Sciences of the United States of America, 115(12)

ISSN

0027-8424

Authors

Kraay, Alicia NM Brouwer, Andrew F Lin, Nan <u>et al.</u>

Publication Date 2018-03-20

DOI 10.1073/pnas.1719579115

Copyright Information

This work is made available under the terms of a Creative Commons Attribution-NonCommercial-NoDerivatives License, available at <u>https://creativecommons.org/licenses/by-nc-nd/4.0/</u>

Peer reviewed

Modeling environmentally-mediated rotavirus transmission: the role of temperature and hydrologic factors

Alicia N.M.Kraay^a, Nan Lin^a, Andrew F. Brouwer^a, Philip A. Collender^b, Justin V. Remais^b, and Joseph N.S. Eisenberg^{a, 1}

^aDepartment of Epidemiology, University of Michigan, 1415 Washington Heights, Ann Arbor, MI 48104; ^bDepartment of Environmental Health Sciences, University of California, 2199 Addison St, Berkeley, CA 94720

This manuscript was compiled on January 18, 2018

Rotavirus is considered a directly transmitted disease due to its high infectivity. Environmental pathways have, therefore, largely been ignored. Rotavirus, however, persists in water sources, and both its surface water concentrations and infection incidence vary with temperature. Here, we examine the potential for rotavirus to exploit water as a transmission pathway. We employ a mechanistic model that incorporates both direct and waterborne transmission pathways, coupled with a hydrological model, and we simulate rotavirus transmission between two communities with interconnected water sources. To parameterize temperature dependency, we estimated temperaturedependent decay rates in water through a meta-analysis. Our metaanalysis suggests that rotavirus decay rates are positively associated with temperature (N=39, p<0.001). This association is stronger at higher temperatures (over 20°C), consistent with tropical climate conditions. Our model analysis demonstrates that water could disseminate rotavirus between the two communities for all modeled temperatures. While direct transmission was important for disease amplification within communities, waterborne transmission could also amplify transmission. In standing water systems, the modeled increase in decay led to decreased disease, with every 1°C increase in temperature leading to up to a 2.4% decrease in incidence. These effect sizes are consistent with prior meta-analyses, suggesting that environmental transmission through water sources may partially explain the observed associations between temperature and rotavirus incidence. Waterborne rotavirus transmission is likely most important in cooler seasons and in communities that use slow-moving or stagnant water sources. Even when indirect transmission through water cannot sustain outbreaks, it can seed outbreaks that are maintained by high direct transmission rates.

Rotavirus | Environment | Temperature

iarrheal disease is the fifth leading cause of mortality worldwide and the third leading cause of death among children under five. The World Health Organization estimated that diarrheal disease accounted for 2.9% of the Disability Adjusted Life Years (DALY) burden worldwide in 2015, making it the sixth leading contributor to the burden of disease (1). A recent multi-country study on diarrheal disease etiology showed that rotavirus was one of four pathogens that together accounted for the vast majority of severe diarrheal disease (2). The recent introduction of rotavirus vaccines has reduced the prevalence of severe cases but is less likely to impact the overall transmission rate, especially given variable coverage of the rotavirus vaccine in low- and middle-income countries, as well as lower vaccine efficacy in the developing world (3, 4). As such, rotavirus is likely to remain an important pathogen for the foreseeable future.

Like other enteric pathogens, rotavirus can be spread either directly from person to person or through contact with contaminated water. However, because of its low infectious dose and high shedding rate among infected individuals, environmental control is generally not considered effective for rotavirus (5) and indirect transmission through water is often not considered when analyzing rotavirus transmission dynamics (6–10). Instead, researchers have generally assumed that transmission through contaminated water is negligible in comparison with transmission occurring within homes and from person to person (with fomites as an intermediary).

In the tropics, meta-analyses have estimated that every 1°C increase in temperature is associated with a 4–10% decrease in rotavirus-associated diarrheal disease incidence (11, 12). Given that there is some evidence of publication bias for the temperature rotavirus effect and that other seasonal processes may affect risk, the true association may be closer to the lower end of this range (12). In temperate regions, researchers have seen that rotavirus is more common during the cooler weather months (13). A number of studies have examined fomite transmission, and although rotavirus can survive on fomites for long periods of time (on the order of weeks), there is a lack

Significance Statement

Although rotavirus persists in water and its incidence varies with temperature, the potential role of waterborne transmission to explain this seasonal pattern in low income countries has thus far been neglected. We find that water can affect the size of outbreaks for larger communities that draw water from slow-moving or stagnant sources and is likely most important in cooler seasons. Water can also spread outbreaks between communities, even when environmental transmission cannot sustain outbreaks. Our model may help explain temperatureincidence associations found in prior meta-analyses in the tropics, where humidity is higher year-round, and shows that the effect of temperature on risk depends on local hydrologic conditions. These findings provide insight into why the effect of temperature might be context-specific.

A.N.M.K. conceived of the study and analytic approach, carried out all analyses, and drafted the manuscript; N.L. performed the literature review, helped conduct the meta-analysis, and helped draft the manuscript; A.F.B. helped design the transmission model analysis and helped draft the manuscript; P.A.C. helped design the meta-analysis approach and helped draft the manuscript; J.V.R. helped design the meta-analysis approach and helped draft the manuscript; J.N.R. belped design the meta-analysis approach and helped draft the manuscript; J.N.R. belped design the meta-analysis approach and helped draft the manuscript; J.N.R. belped design the meta-analysis approach and helped draft the manuscript; J.N.S.E. con-

We have no conflicts of interest to report.

¹Corresponding Author: Department of Epidemiology, University of Michigan, 1415 Washington Heights, Ann Arbor, MI 48104 (email: jnse/@umich.edu) of evidence that temperature has a strong effect on rotavirus persistence on fomites (14-18). In low- and middle-income countries in tropical settings, an alternative environmental pathway, such as water, may contribute to the temperature-driven incidence patterns. In temperate climates, seasonality in direct transmission is more likely to be explained by large seasonal shifts in relative humidity, with drier conditions promoting more direct transmission (14-18).

Various studies have attempted to predict the mechanisms by which increased temperature might alter risk of diarrheal disease by changing pathogen survival and growth dynamics, pathogen virulence, and human behavioral factors (19). In general, the burden of diarrheal disease is expected to increase as climate change progresses (20); however, there is a great deal of heterogeneity by pathogen (12). For bacterial pathogens, increased temperature is related to decreased persistence but can also lead to expression of genes associated with increased virulence (21, 22), and the net effect of increased temperature is thus expected to increase risk. In contrast, higher temperatures are only expected to decrease environmental persistence of viruses, including rotavirus (19), and thus are expected to reduce viral infection risks (11). Although this body of work has helped establish the associations between environmental factors and disease transmission risk, studies are needed to better elucidate the mechanisms by which environmental changes, such as increased temperature, affect risk (23).

In this study, we use a mechanistic model of rotavirus transmission to evaluate the potential of rotavirus to be transmitted through the environment. To parameterize this model and to assess how temperature might impact this pathway, we conduct a meta-analysis of experimental data to derive a statistical model for the relationship between temperature and rotavirus persistence in water. We then apply the resulting temperature sensitivity function within our rotavirus transmission model in two communities linked by a flowing body of water, examining the potential for waterborne transmission to disseminate rotavirus infections between communities and amplify cumulative incidence within communities. The hydrologic conditions in these two communities were parameterized to reflect the conditions found in our study site in rural, coastal Ecuador (24), which encompasses three rivers with distinct average flow velocities: the Onzole (17 km/day), the Cavapas (26 km/day), and the Santiago (69 km/day) (25). We also assess how changing temperature might impact risk at both the community and regional scales and explore how local hydrologic conditions and access to safe water and sanitation services could impact this relationship. The aim of this analysis is to estimate how temperature-related changes in waterborne transmission, mediated by the waterborne decay rate, might help explain correlations between rotavirus incidence and temperature or season, especially in tropical settings.

Results

The Basic Reproduction Number. Using the next-generation approach, we derived the following expression for \mathcal{R}_0 in one community (see supplement):

$$R_0 = \frac{\beta_H}{\gamma} + \frac{0.5\beta_W N \rho \frac{(1-c)\phi}{V}}{\gamma(\mu+\nu)}$$
[1]

$$=R_{0,H}+R_{0,W}$$
 [2]

In equation 1, β_H is the transmission rate for direct, humanto-human transmission, γ is the recovery rate, β_W is the transmission rate from water, N is the population size, ϕ is the shedding rate for infected individuals, ρ is the water consumed per day, 1-c is the fraction of shed pathogens that ultimately mix in the local water supply, μ is the temperaturedependent rotavirus decay rate in water sources, and ν is the river flow velocity. See Table 2 for more details about the parameters. The basic reproduction number can be written as a sum of two submodel reproduction numbers (equation 2) $\mathcal{R}_{0,H}$ (the \mathcal{R}_0 from human-to-human transmission) and $\mathcal{R}_{0,W}$ (the \mathcal{R}_0 for water-to-human transmission).

This expression leads to several insights. First, when all transmission is person-to-person ($\beta_W = 0$), the reproduction number reduces to the familiar SIR form of β/γ , which is the general formulation assumed by other rotavirus transmission models. Second, for water-to-human transmission, the river flow velocity impacts how strong of an effect temperature has on $\mathcal{R}_{0,W}$. For flowing water systems ($\nu > 0$) temperature related die-off $(\mu = K(T))$ has a negligible effect whereas in standing water ($\nu = 0$), temperature related die off becomes more important for determining transmission potential. Third, the river flow velocity also impacts persistence in the local water supply. In flowing systems, the speed of flow is generally much faster than the speed of pathogen decay. As such, the length of pathogen persistence in the local water supply is much more strongly affected by hydrological flow dynamics. To facilitate comparison between these two rates for the dissemination analysis, we fixed the distance of each river compartment, scaling each compartment by its length (see supplement for details). Thus, flow velocity and decay rates are on the same scale in our model (units of day^{-1}). In standing water systems, the length of pathogen persistence in the environment is entirely driven by temperature. Even in standing water, however, the most important contributor to water-human transmission $(\mathcal{R}_{0,W})$ is the size of the water reservoir (V) relative to the effective shedding rate $(1-c)\phi$. Overall, the size of the water reservoir relative to the human population is the critical driver of the strength of water-tohuman transmission, as it determines the degree of dilution that occurs when pathogens are shed into the water.

Meta-Analysis: Temperature and Die Off. Our literature search returned 57 articles. Of these articles, we identified 9 that met our inclusion criteria containing a total of 39 experiments for use in our analysis. These studies are summarized in Table 1. We used data from these 39 experiments to estimate the dieoff rate (day^{-1}) for each study. The decay rates ranged from 0.008 to 0.996 per day, with a median decay rate of 0.056/day (corresponding to a median survival time of 18 days).

Our pooled data across all 39 experiments were fit by general linear regression,

$$\log(K(T)) = \beta_0 + \beta_1(T - 25) + \beta_2 \chi_{\text{nat}}, \quad [3]$$

and support the hypothesis that the decay rate K(T) of rotavirus is temperature (T) dependent, with higher temperature

Table 1. Studies included in the meta-analysis. The phrase 'combined' is used for simple aqueous solutions at a neutral pH, such as phosphate-buffered saline (PBS).The phrase 'purified' is used for sterile, distilled water. The phrase 'treated' is used for water taken from raw water or sewage that was treated to remove microbes prior to analysis. More details about the estimated decay rates from each experiment are in the supplement.

Author, Year	No. Experiments	Temperature Media Type		Ref
Espinosa et al., 2008	4	15, 25°C	Natural, Combined	(26)
Ward et al., 1986	9	4, 16, 23, 27, 29, 37° C	Natural	(<mark>27</mark>)
Raphael et al., 1985	12	4, 20°C	Natural, Treated	(<mark>28</mark>)
El-Sanousy et al., 2014	3	4, 22, 35° C	Natural	(<mark>29</mark>)
Hansen et al., 2007	1	22° C	Treated	(<mark>30</mark>)
Sattar et al., 1984	2	4, 20°C	Treated	(31)
Höglund et al., 2002	2	5, 20°C	Combined	(<mark>32</mark>)
Chung et al., 1993	2	5, 25°C	Natural	(<mark>33</mark>)
McDaniels et al., 1983	4	8, 26°C	Purified, Treated	(<mark>34</mark>)

leading to increased die-off ($\beta_1=0.078$, SE=0.016, p < 0.01) In our model, $\exp(\beta_0)$ is die-off rate at 25°C, and $\exp(\beta_1)$ quantifies the multiplicative change in decay rate per 1°C increase in temperature. Due to the exponential relationship between temperature and die-off, the effect of a given change in temperature was small below 20° C (Figure 1). It is known that competition between microbes can influence decay rates and some of these experiments were conducted in natural water sources that could contain other microbes. To mitigate this concern, we also adjusted for this variable, with natural water sources being the only water type likely to contain other microbes. As expected, experiments conducted in natural water sources (quantified using an indicator variable, χ_{nat}) that could contain other microbes had higher decay rates ($\beta_2=1.21$, SE=0.327, p < 0.01). Although the effect estimate for temperature attenuated slightly after adjusting for this possible competition effect, the association remained significant. There was no significant interaction between water type and temperature in our dataset.



Fig. 1. Temperature (T) and decay rate (K(T)) with prediction intervals from the adjusted model. Data, predicted decay rates, and prediction intervals are shown in black for natural water sources and in gray for all other water sources. The predicted decay rates for each temperature are shown in solid lines and the 95% prediction intervals are shown in dashed lines for both types of water sources. For a graph of these decay rates on the linear scale, see supplement.

The adjusted, fitted regression was

$$\log(K(T)) = -2.51 + 0.078(T - 25) + 1.21\chi_{\text{nat}}.$$
 [4]

For our simulations, we set $\chi_{\text{nat}} = 1$. (Comparisons to simulations run with the model unadjusted for water type may be found in the supplement. The results were similar.)

Amplification in Water Sources. In our simulations, for flowing systems, water can only amplify infection when the available volume of drinking water sources is low, because residence times in the local water supply are too short for pathogen concentrations to accumulate locally. However, for standing water, temperature can be an important determinant of community risk at moderate-to-high volumes, with higher temperatures reducing the strength of indirect transmission (Figure 2) and leading to lower cumulative incidence (Figure 3).

The strength of the temperature effect (adjusted by chang- $(ing \mu)$ depends on both the strength of direct person-to-person transmission and the size of the water reservoir, with temperature having a stronger effect when direct transmission is weaker (when β_H is smaller) and the per capita water availability is higher (V is higher). When per capita water availability is 1,000 L, temperature only affected risk at higher temperatures, with an average risk difference per $^{\circ}$ C of 1.0%, 0.3%, and 0% for temperatures between 20°C and 37°C at $\mathcal{R}_{0,H}$ values of 1, 2, and 3 respectively (Figure 3B). When per capita water availability was 10,000 L, temperature had the largest effect on risk at lower temperatures with risk differences of 2.4%, 0.5%, and 0.2% per °C between 10°C and 20°C for $\mathcal{R}_{0,H}$ values of 1, 2, and 3 respectively (Figure 3C). Between 20°C and 37°C, these same risk differences were smaller, with values of 1.9%, 0.3%, and 0.1% respectively.

For standing water at moderate dilution, the amount of sanitation coverage needed to reduce $\mathcal{R}_{0,W}$ below 1 varied by temperature, with higher coverage needed to interrupt waterborne transmission at lower temperatures. When dilution was high, $\mathcal{R}_{0,W}$ was always less than 1 irrespective of the degree of attenuation of shedding. For flowing water, $\mathcal{R}_{0,W}$ was only greater than 1 when dilution was low, and the amount of sanitation coverage needed to reduce $\mathcal{R}_{0,W}$ below 1 depended on the flow velocity, with less coverage needed in faster flowing rivers (Figure 2).

Dissemination through water. In order for an outbreak to move downstream through water sources, we found that $\mathcal{R}_0 > 1$ at the upstream location of outbreak seeding was a sufficient condition for downstream outbreaks, as long as the transmission term from water was positive ($\beta_W > 0$). However, this condition was not necessary when the water was highly concentrated (due to low volume and slow flow rates).

In particular, when direct transmission dominates ($\mathcal{R}_{0,W} \ll \mathcal{R}_{0,H}$), dissemination to the downstream community (i.e. an outbreak at the downstream location) does not occur when the upsteam $\mathcal{R}_0 < 1$, regardless of the initial conditions (Figure 4A). In contrast, when indirect transmission dominates ($\mathcal{R}_{0,H} \ll \mathcal{R}_{0,W}$), outbreaks can occur in the downstream community when the upstream $\mathcal{R}_0 < 1$ if the water volume is low, depending on the initial conditions (Figure 4B). Hence, our model suggests that dissemination of rotavirus along waterways can occur even when local person-to-person transmission conditions are not conducive to outbreaks.

This model behavior is consistent across a wide range of die-off and flow velocity values, including both when there is no direct transmission ($\mathcal{R}_{0,H} = 0$, Figure 5) and when both pathways were active ($\mathcal{R}_{0,H} > 0$ and $\mathcal{R}_{0,W} > 0$ but $\mathcal{R}_{0,H} + \mathcal{R}_{0,W} < 1$). In this last condition, the $\mathcal{R}_0 = 1$ threshold occurs at a higher flow rate whose value is proportional to the



Fig. 2. The strength of indirect, water-human, transmission ($\mathcal{R}_{0,W}$) for A) flowing water ($\nu > 0$) at low volume (10^5 L), B) standing water ($\nu = 0$) at moderate volume (1,000 L/person, 10^6 L total volume), and C) standing water ($\nu = 0$) at moderately-high volume (10,000 L/person, 10^7 L total volume). For flowing water, the flow velocity (ν) is more important than decay rates, so the sensitivity analysis is shown as a function of flow rates (ν) and attenuation due to water, sanitation, and hygiene interventions and natural processes (c). For standing water, the water is not flowing so decay rates (μ) and attenuation of shedding (c) become important for waterborne disease dynamics. For flowing water, the pathogen decay rate was set to 0.056, the median decay rate found in the studies included in our meta-analysis.

strength of direct transmission (Figure 5). At higher water source volumes above 100,000 L (100 L/person), there is no dissemination to the downstream community when the local \mathcal{R}_0 is less than 1.

One condition that might prevent dissemination is if rotavirus dies off faster than it is transported. In our simulations, higher die-off rates caused by higher temperature were not sufficient to prevent dissemination through waterways because the die-off rates from our meta-analysis, ranging from 0.008 to 0.996/day (corresponding to mean survival times from 1–125 days), were lower than the river flow rates we observed in our Ecuador field site. Therefore, in practice, the parameter region where die-off rates are higher than flow rates probably only occurs for very slow flowing river systems where flow rates are less than 1 km/day. Furthermore, our simulations suggest that even when flow rates are slower than 1 km/day, dissemination may occur if volume is low.

In summary, our simulations suggest that a sufficient condition for dissemination is that, locally, $\mathcal{R}_0 > 1$ but that there is a narrow parameter regime when the local, upstream \mathcal{R}_0 is less than 1 but where dissemination can also occur under low volume conditions.

Discussion

We find that environmental transmission of rotavirus through water sources may be an important source of risk in tropical climates at both the community scale (due to amplification) and regional scale (due to dissemination). The role of the environment in rotavirus transmission was affected by temperature; river flow velocity; water, sanitation and hygiene interventions; and the size of the water reservoir (which determines dilution). Increases in temperature were found to be associated with exponential increases in the decay rate of rotavirus in water sources. We would therefore expect temperature to only appreciably increase die-off in tropical locations where the water temperatures can easily exceed 20°C. However, even at lower temperatures small changes in ambient temperatures can have a larger impact on risk for communities drawing water from more dilute surfaces, such as lakes. Communities linked by flowing rivers could disseminate infection but their water supplies did not amplify within-community outbreaks unless dilution was low, whereas outbreaks in communities with standing water sources could be amplified by environmentally mediated transmission when dilution was low-to-moderate. In the real world, these patterns may help to describe distinct disease dynamics among communities using rivers-as opposed to ponds and lakes-for water sources, as well as communities with riverine water sources with time-varying flow velocity and connectivity. Furthermore, even in communities served by flowing surface water sources, pathogens may accumulate in areas of slow flow along the river's edge (25). Disease dynamics in communities where these zones serve as foci for interactions between the human population and the water source may exhibit qualities of both flowing and standing water communities as described by our model.

Prior researchers have attempted to understand how rotavirus may be influenced by temperature and other seasonal factors using both time series analyses and meta-analysis techniques. In general, these studies have found that rotavirus infections are more common in cooler seasons with average effects of 4–10% decrease in incidence per 1°C increase in temperature (11, 12). In our model, every 1°C increase in temperature leads to a total 0.1–2.4% decrease in incidence for standing water sources, which is at the lower end of the average associations between temperature and rotavirus seen in prior meta-analyses. There are a number of reasons that our results might not conform exactly to the results of these prior time series studies. First, because the relationship between temper-



Fig. 3. Cumulative incidence (fraction) for different temperatures and strength of direct transmission. $\mathcal{R}_{0,H}$ values of 1, 2, and 3 are shown in solid, dotted and dashed lines respectively for A) standing water ($\nu = 0$) at low volume (10^5 L), B) standing water ($\nu = 0$) at moderate volume (1,000 L/person, 10^6 L total volume), and C) standing water ($\nu = 0$) at moderately-high volume (10,000 L/person, 10^7 L total volume). For all three panels, attenuation of shedding due to water, sanitation, and hygiene interventions and natural processes (*c*) is set to 0.9. The maximum effect of temperature occurs at this level of attenuation and high dilution, with $\mathcal{R}_{0,W}$ decreasing from over 0.98 to 0.11 as temperature increases from 10 to 37° C for standing water. For flowing water, temperature has no effect on cumulative incidence for any volume because flow is much faster than decay and $\mathcal{R}_{0,W}$ only changes from 0.05 to 0.04 as temperature increases from 10°C to 37°C.

ature and decay is exponential, the effect of any temperature change depends on the baseline temperature. This means that larger changes in pathogen survival are expected in areas that are already hotter on average, such as the tropics. Second, many of the studies considered in these meta-analyses may also be subject to confounding by other unmeasured factors including differences in local hydrological conditions. Thus other seasonal processes are likely to explain much of the remaining temperature association, and the average associations in the literature may be inflated by publication bias (12).

The impact of temperature on disease transmission is likely to be different in temperate compared with tropical climates. In temperate regions, many factors in addition to temperature change seasonally that could cause a large shift in the strength of transmission, including seasonal changes in humidity. Thus, rotavirus seasonality in temperate climates is likely to be multifactorial and the result of many interdependent processes, which we do not explore here. While humidity has a large impact on the decay rate of rotavirus on fomites, temperature has a much smaller impact. The majority of the literature suggests that at high humidity conditions such as those found in the tropics, rotavirus survival on surfaces is likely to be long enough to sustain transmission for all temperatures (14–18). Thus, while direct transmission may have a seasonal component, particularly in temperate climates, it is unlikely that the seasonality of rotavirus transmission in the tropics can be explained by temperature-related variations in fomite-mediated transmission alone (14-18). A detailed analysis of seasonal changes in transmission parameters for both pathways is beyond the scope of this paper, but our results do suggest that temperature, and its impact on waterborne transmission, might impact seasonality in rotavirus transmission, particularly in the tropics.

Although our meta-analysis showed that higher temperatures were associated with faster decay, our model predicts that this effect of temperature would not be sufficient to prevent dissemination through water sources in sites similar to our study site in Ecuador. For communities with standing water sources, water also acted as an amplifier of local transmission. Even though the degree of coupling of infectious pathogens between the water supply and infectious people is probably lower in large cities that use standing water sources, the sensitivity of our model to the per capita water availability suggests that larger cities with poor sanitation infrastructure where most residents get their drinking water from a single reservoir may be especially vulnerable to waterborne transmission. For communities using standing water sources, the effect of temperature was more pronounced, and higher temperatures led to smaller outbreaks due to the decreased persistence of rotavirus in the water supply.

Dissemination of rotavirus through flowing waterways. Together, our results suggest that water is likely to be a sufficient disseminator of infection in most flowing systems. Although it is not necessary for the overall reproduction number in the upstream community to be greater than 1 for dissemination to occur, in most contexts the sufficient condition $(\mathcal{R}_0 > 1)$ is likely to be met either through direct transmission alone $(\mathcal{R}_{0,H} > 1)$ or indirect transmission if there is increased contamination levels due to low water volumes $(\mathcal{R}_{0,W} > 1)$. Because the overall reproduction number is a function of both direct and indirect transmission, it is sufficient for the sum of both submodel \mathcal{R}_0 's to be greater than 1. The submodel reproduction number $\mathcal{R}_{0,H}$ is likely to be greater than 1 in most contexts in the absence of focused interventions targeting this pathway, both because the shedding rate of rotavirus



Fig. 4. The effect of increasing the initial upstream outbreak size on cumulative incidence in the downstream community when A) Most transmission occurs directly from human to human and the submodel reproduction number for water was <0.01, and B) All transmission occurs through the water and direct transmission is set to 0. This result only occurs at low volumes (100,000 L).



Fig. 5. Location of outbreak (downstream and/or upstream community) as a function of pathogen decay rates and effective flow velocity. An outbreak is defined by a cumulative incidence of at least 5% after 365 days. The solid line corresponds to the situation when $\mathcal{R}_{0,W} = 1$. Above this line, $\mathcal{R}_{0,W} < 1$ and below this line $\mathcal{R}_{0,W} > 1$. For these simulations, the sub-model reproduction number for direct transmission was set to 0 and volume was set to 100,000 L.

is extremely high relative to its median infectious dose and because rotavirus is persistent on a variety of environmental surfaces (14–18). When $\mathcal{R}_{0,H} > 1$, the sufficient condition for dissemination will always be met, as long as water remains sufficient to disseminate enough pathogens to cause an outbreak in downstream communities. Even when direct transmission is not sufficiently high, low water volumes (increasing indirect transmission) or high initial outbreak sizes (increasing the probability of a breakthrough transmission event) may allow water to disseminate infection between communities.

Multiple pathways: direct and indirect transmission. While some seasonality patterns in rotavirus incidence may be partially explained by dynamic resonance related to the birth rate, many other factors may underlie this typical seasonal pattern of risk (8, 9, 35), including environmental variables like temperature. Because rotavirus is highly infectious, direct transmission alone may be sufficient to cause nearly universal infection of the susceptible population. However, in this analysis we have identified situations in which indirect transmission can affect the spread of disease through a watershed and may also be sufficient to cause large outbreaks. When two sufficient causes (i.e., environmental and direct transmission) are both met in a population (36), conventional epidemiologic studies that analyze case data alone may not detect this environmental transmission because multiple mechanisms are capable of explaining the observed dynamics (2, 6, 37, 38). By explicitly considering this environmental route of transmission, we were able to better explore the potential importance of both pathways. Our model shows that while direct human-tohuman transmission is likely to be more important for local amplification, indirect transmission may be sufficient to both sustain and spread outbreaks. Furthermore, if one assumes uninterrupted flow and surface water availability, waterborne transmission appears to be more important in cooler and drier seasons, when rotavirus persistence in the environment is high and the dilution effect is weak. In settings where certain assumptions of our model hold, i.e., that accumulation of and exposure to waterborne rotavirus can be adequately represented with a single water compartment and where access to safe water and sanitation is poor, inland cities with poor sanitation access are also expected to be especially vulnerable to waterborne transmission because of their relatively lower per capita water availability. In these settings, interventions like vaccination that impact both direct transmission and indirect pathways may be needed to reduce risk of infection within communities and to stop the spread of infection through watersheds.

Our model shows that the degree of intervention needed to interrupt water transmission depends on both local hydrologic conditions and ambient temperature. For standing water systems, less effort may need to be invested to prevent the accumulation of viruses in local water bodies when ambient temperature is high. For communities that utilize flowing surface water systems, targeting direct transmission alone may be sufficient to prevent disease spread as long as the size of the water reservoir is relatively large. When the size of the water reservoir is relatively small, the degree of intervention needed in flowing systems depends on the speed of water flow. Furthermore, if water is in fact an effective disseminator of rotavirus, restricting person-to-person contact may be sufficient to prevent the spread of rotavirus between communities for systems with greater per-capita water availability. In other types of communities, direct human-to-human transmission may be important for local amplification, but indirect transmission may sustain and spread outbreaks. This information can help public health practitioners identify optimal approaches to decrease environmentally mediated rotavirus risks in resource-limited settings by considering local climatological and hydrological conditions.

We acknowledge that many questions remain about how rotavirus is transmitted: there is strong evidence both that the overall reproduction number \mathcal{R}_0 is high and that the environment can be important for transmission. In fact, the processes underlying rotavirus transmission are likely to be highly complex, with the potential for super-spreading and large heterogeneities in transmission by age (6). In the United States, where drinking water and sanitation facilities are generally safe, rotavirus was still a major source of disease risk prior to the introduction of the vaccine, suggesting that water and sanitation interventions alone are unlikely to be adequate to interrupt transmission (39). Because water remained an important source of risk in our model for communities resembling larger cities even when shedding was highly attenuated, our results are consistent with this line of research. Future work combining environmental sampling over multiple seasons with case data would be particularly useful to help validate our conclusions. Genetic data could also help clarify the relative importance of different transmission routes. In addition, future studies on the effect of humidity on rotavirus persistence on fomites could help clarify other sources of seasonality in direct transmission.

Implications for trends in rotavirus risk under climate change.

Given that global temperatures are projected to increase as climate change proceeds (40), many researchers have attempted to understand the relationship between temperature and risk of disease. In our model, we showed that one reason for the observed decrease in rotavirus infections at higher temperatures might be increased die-off of rotavirus in water sources. Other pathogens, however, have different temperature signatures, and many studies have shown that risk of bacterial and protozoan diarrhea is, in fact, slightly elevated at higher temperatures (12). However, these associations do not necessarily preclude increased die-off at higher temperatures for bacteria, but rather reflect the net balance of factors for bacteria compared with viral pathogens. For example, for many bacterial species, temperature activated virulence genes have been identified which may lead to increased risk of disease at higher temperatures even in the presence of faster die off (21, 22). Some bacteria may also have higher replication rates at higher temperatures (12). One implication, then, of our results is that the etiology of waterborne diarrhea cases might shift away from rotavirus and towards bacterial and protozoal infections as global mean temperatures continue to increase. On the other hand, rotavirus may be able to adapt to slowly increasing temperature, suggesting the need to study the temperature sensitivity of rotaviruses from different climates (41). Another potential implication is that the waterborne transmission route for rotavirus may become less important as the average temperature increases, even if overall transmission does not decline. Given the non-linear relationship between temperature and die-off, this shift is more likely in the tropics.

Conclusions. While some researchers acknowledge that waterborne transmission of rotavirus can occur, its importance is often downplayed relative to direct transmission (5). Our model analysis reaffirms that direct transmission likely plays a central role in amplifying rotavirus infection within communities. We used a transmission model parameterized using temperature-specific rotavirus die-off data to show that water is sufficient to both disseminate rotavirus infection between hydrologically connected communities, initiating local transmission cycles, and to amplify transmission under conditions of relative water scarcity in flowing systems and in standing water for a wide range of water availability. Our model indicates these environmental effects are most important in cooler seasons and, may be particularly important in larger cities with poor sanitation. These findings highlight the need to consider waterborne pathways in rotavirus control interventions, particularly in endemic regions in the tropics where the accumulation of virus in waterways may be more pronounced.

Materials and Methods

An overall diagram of our conceptual framework and the flow of analysis is shown in Figure 6 below (adapted from (19)). We built an ordinary differential equation (ODE) transmission model to examine the role of temperature on rotavirus transmission patterns. To parameterize this model and to assess the effect of temperature on this waterborne pathway, we conducted a meta-analysis of studies that provided data on rotavirus die-off rates in water and developed a statistical model relating temperature to die-off rate. We then used this model to assess the role of water in dissemination of rotavirus between communities and amplification of rotavirus infection cycles within communities. All analyses were conducted in R (v. 3.1.2). We integrated the ODE model using the deSolve package.

Transmission Model Structure. To explore the potential for hydrological transmission of rotavirus infections between sites, as well as the climate sensitivity of such an epidemiological system, we developed a susceptible, infectious, recovered (SIR) model representing two hydrologically-connected communities at either end of



Fig. 6. Conceptual framework of the analysis. The meta-analysis portion of our study informs arrow A, whereas the transmission model analysis informs arrows B and C. The transmission model also considered the role of the size of the water reservoir, which may vary based on the type of water source a community uses. This figure was adapted from (19).

a 10 km river reach. In our model, susceptible, infectious, and recovered individuals in community *i* are given by compartments S_i , I_i , and R_i , and transmission can be either direct—from contacts between susceptible and infectious persons, which occur at rate β_H —or indirect—from ingestion of pathogens persisting in the aqueous environment (with contact rate β_W and per contact volume ingested ρ). The water compartments (W), described in more detail below, model the concentration of pathogens in the water. We assume that individuals within communities draw water from only one water source, such that the concentration in the water compartment is a reasonable proxy for exposure for all individuals within that communities.

We assume that infectious individuals shed pathogens into the environment at daily rate ϕ , but that the number of shed pathogens is attenuated prior to entering the water compartment by a combination of natural environmental decay and local sanitation and hygiene interventions (removing a fraction of pathogens c). In communities with built infrastructure, this parameter c reflects not only processes relating to runoff and decay in the environment, but also processes related to wastewater treatment, and thus most likely has a higher value. Larger cities may also have higher levels of attenuation due to their more complex topology. Upon deposition in the water, pathogens are diluted based on the total volume of the community's water source (V_i) . Prior researchers have suggested that the per capita volume of water resources available may be smaller for large settlements with built infrastructure than for smaller communities. like those modeled here, where the size of the water reservoir is determined by natural rainfall and surface water flows, which often leads to greater dilution of shed pathogens (42). We thus simulated across a range of available volumes by adjusting to produce water reservoir sizes that were consistent with a small pond (low dilution) to a medium-sized lake (high dilution) (43) (see supplement for additional details). The low dilution scenario also approximates the minimum the per capita rate of water availability targeted for larger cities with piped water (44). Using our meta-analysis (procedures described below), we assume that the concentration of rotavirus in the water attenuates as a first order reaction with rate $\mu = K(T)$ (eq. (Eq. (3)). We model infections produced by ingestion of contaminated water using a linear dose-response function (45), which we deemed adequate because the pathogen doses in our simulations fell within the linear range of the typical dose-response curves (46).

We modeled movement of rotavirus from upstream to downstream communities using a mixing cell hydrological transport model (47). In brief, this model approximates the advectiondiffusion equation for solute transport, modeling downstream flow through a series of compartments representing river reaches. While no diffusion of pathogens occurs in the upstream direction from point source inputs, the method produces a moving distribution of pathogen concentrations, with mean position at time after entering the system determined by the velocity of advection, and variance in pathogen position at time, i.e. rate of diffusion, proportional to the product of the advection velocity and cell length (47). Thus, flowing systems and reasonably large spatial scales (i.e. where little diffusion in the upstream direction of pathogen sources is expected), our mixing cell approach closely approximates a solution to the advection-diffusion equation. In our model, we used a total of 11 compartments (one for each village and nine intermediary), each representing a 1 km stretch of river, between which the rate of flow is determined by the flow velocity, which is given by ν . We assume that there is no sedimentation of pathogen between communities given the small virus particle size and the length of the river reach between communities. However, attenuation due to sedimentation could be approximated by assuming a slightly lower flow velocity. Pathogens entering the water at the upstream community are allowed to either move to the next compartment or decay, and no pathogens can be shed into the water between villages. More details about the mixing cell model, including comparisons with other methods, is in the supplement.

The model diagram is shown in Figure 7, and the model equations are given below. Parameters used in simulations and their values are shown in Table 2.

$$\begin{split} \dot{S}_{i} &= -\frac{\beta_{H}}{N} S_{i} I_{i} - \frac{\beta_{W} \rho}{2} S_{i} W_{i}, \\ \dot{I}_{i} &= \frac{\beta_{H}}{N} S_{i} I_{i} + \frac{\beta_{W} \rho}{2} S_{i} W_{i} - \gamma I_{i} \\ \dot{R}_{i} &= \gamma I_{i}, \\ \dot{W}_{1} &= \frac{(1-c)\phi}{V_{1}} I_{i} - (\mu + \nu) W_{1}, \\ \dot{W}_{0}^{\text{int}} &= W_{1}, \\ \dot{W}_{j}^{\text{int}} &= \nu W_{j-1}^{\text{int}} - (\mu + \nu) W_{j}^{\text{int}}, \quad 1 \le j \le 9, \\ \dot{W}_{2} &= \frac{(1-c)\phi}{V_{2}} I_{2} + \nu W_{9}^{\text{int}} - (\mu + \nu) W_{2}, \end{split}$$

$$[5]$$

The transmission rate parameters β_W and β_H and the recovery parameter γ are assumed to be the same for both communities. In this simulation, the two communities have the same population size and available water volume. Flow velocities between the two sites are reasonable approximations of the hydrological characteristics of our study site in rural, coastal Ecuador (24), which encompasses three rivers with distinct average flow velocities: the Onzole (17 km/day), the Cayapas (26 km/day), and the Santiago (69 km/day) (25).



Fig. 7. Rotavirus transmission model diagram. Dashed arrows represent transmission events and solid arrows represent transition of people or pathogens between model compartments. Villagers and city residents are classified as susceptible (S), infected (I) and Recovered (R). We also track contamination in the village water source (W). Susceptible people can become infected by either direct person to person transmission or indirect person-water-person transmission within their own community. Pathogens in the local water source. Pathogens in water sources can also decay in each community or between communities. The model parameters are defined in Table 2.

Table 2. Parameter estimates, units, and their sources. Water volume calculations and details regarding the scaling of flow velocity are shown in the supplement.

Parameter	Symbol	Units	Value	Source
Transmission rate from human to human	β_H	1/day	[0, 0.6]	Varied
Transmission rate from water to human	β_W	water contact/day	3	(48)
Recovery rate	γ	1/day	0.2	(49)
Median infectious dose	n_{50}	ffu	10	(7)
Shedding rate	ϕ	n_{50} /person/day	2,605	(7, 50-52)
Attenuation of shedding	c	-	[0, 1]	Varied
Water consumed per day	ρ	volume/water contact	0.4	(48)
Population size	N	people	1000	Assumption
Water volume	V_i	L	$[10^5, 10^9]$	Varied
Pathogen decay rate	μ	1/day	[0,1]	K(T)
Flow velocity	ν	km/day	[0, 69]	(25)

Literature Review and Meta-Analysis.

Data Collection. To parameterize our transmission model, we retrieved a comprehensive list of studies that estimate rotavirus dieoff in water samples at known temperatures by querying PubMed, Scopus, and Web of Science with the following search string:

(rotavirus OR 'rotavira^{*}') AND (temperature OR heat) AND ('die off' OR inactivation OR dieoff OR growth OR survival) AND (water OR aqueous OR aquatic OR marine).

Studies were included that: 1) reported results from at least one rotavirus experiment; 2) that took place in purified or treated water (i.e., reverse osmosis water, sterile distilled water, tap water, etc.), natural water (e.g., from lakes, rivers, creeks, etc.), posttreatment (i.e., tap water, filtered water, etc.), or simple aqueous solutions at neutral pH (i.e., phosphate-buffered saline (PBS)); 3) reported a steady water temperature that did not fluctuate more than 5°C during the course of the experiment; and 4) provided a clear explanation of methods used to assay rotavirus concentration over time. Studies conducted in more complex media (e.g., that added natural organic matter, humid sludges, etc. to aqueous media) were examined for control experiments conducted in pure water but were otherwise excluded. Also excluded were studies that reported only the time required for total inactivation of the rotavirus population.

Temperatures reported in die-off experiments were recorded directly from reviewed studies. Where mean rotavirus concentration was available in a figure or plot, data were extracted using DataThief software (53). Additional information regarding rotavirus serotypes, water source, water treatment, initial viral concentration was also recorded.

Statistical Analysis. We determined the relationship between temperature and die-off rates in two steps. First, we estimated the die-off rate for each experiment. Second, we estimated the relationship between die-off rates in each study and the corresponding temperatures.

The rate of change in viral concentration is commonly modeled as a simple first-order decay process where N(t) is the pathogen population at time t; N_0 is the initial population; and K is the first-order rate constant (54):

$$N(t) = N_0 \times e^{-Kt}$$
^[6]

We used this framework to model pathogen decay in our analysis. We also examined the data for evidence of tailing off, or biphasic decay (55), but few experiments appeared to exhibit biphasic decay and those that did lacked sufficient observations for us to estimate the parameters. We therefore assumed that the true pattern of decay was monophasic for all studies.

After obtaining an estimate of K in each experiment, we calculated the relationship between temperature and the decay rate of rotavirus in aqueous environments. We modeled the decay rate K of rotavirus as a non-linear function of temperature (T) (Eq. (3)) and fit this model using generalized least squares, following prior published approaches used for other pathogens.

Transmission Model Analysis. All model simulations were seeded with one infectious case in the upstream community and were run for 365 days. Both communities were assumed to have a population of 1,000. To better understand the threshold and equilibrium features of our model, we calculated the basic reproduction number, \mathcal{R}_0 , for the local communities using the next generation method (56). Briefly, \mathcal{R}_0 captures the expected number of secondary cases produced by one case during its infectious period after being introduced into a completely susceptible population. If this number is greater than 1 (e.g., greater than replacement), an outbreak seeded near the disease-free equilibrium will transiently grow, and, if it is less than 1, such an outbreak will die out.

The spatially local \mathcal{R}_0 we derive does not necessarily control the outbreak dynamics on the whole system, nor does it precisely control whether an outbreak will occur in the downstream community if seeded in the upstream community. In a similar model, Gatto et al showed that for pathogen invasion to occur along a network it is necessary that the basic reproduction number for at least one of the communities in the network be greater than 1 (57). Hence, we use the spatially local \mathcal{R}_0 for the upstream community as a dynamically relevant and intuitive benchmark for our simulation studies.

Amplification. To determine if water could amplify infection by changing the size of the outbreak, we estimated the overall attack ratio, that is, the cumulative incidence, as a function of the strength of each pathway and estimated how cumulative incidence changed as temperature changes impacted rotavirus decay rates. We also assessed how local hydrologic conditions and attenuation of shedding might impact risk (defined by cumulative incidence at the end of the simulation) both overall and from water sources alone.

Dissemination. We defined dissemination as 5% of the population in the downstream community being infected by the end of the simulation (365 days). We investigated how dissemination might depend on temperature, river flow velocity, and dilution for biologically plausible parameter values (see table 2 for credible parameter ranges). Because sub-threshold outbreaks can occur even when $\mathcal{R}_0 < 1$ (56) and knowing the parameters is alone is not always sufficient to predict whether or not an outbreak will occur, we also analyzed whether or not downstream outbreaks occurred in our model for biologically plausible changes in initial conditions (initial outbreak sizes up to 10%).

We recognize that human travel is also an important disseminator for rotavirus and is needed to explain spread of disease to upstream communities that would not otherwise become infected through flowing waterways. However, in this analysis, we aimed to determine if water was also a sufficient disseminator to downstream communities. Thus, we acknowledge that waterborne dissemination is not necessary to spread disease between communities but aimed to determine if it was also a sufficient pathway.

For both dissemination and amplification analyses, we used the fitted model equation from the meta-analysis to calculate decay rates at different temperatures, simulating across the range of decay rates seen in the literature.

ACKNOWLEDGMENTS. A.N.M.K., N.L.,A.F.B., and J.N.S.E. were supported by the National Institutes of Health (grant R01-AI050038) and NIH MIDAS (grant U01GM110712). P.A.C. and J.V.R. were supported by National Institutes of Health (grant R01TW010286, R01AI125842 and K01AI091864), by the National Science Foundation Water, Sustainability and Climate (grants 1360330 and 1646708), and by the University of California Multicampus Research Programs and Initiatives award MRP-17-446315.

- Kassebaum NJ, M. A, et al. BRM (2016) Global, regional, and national disability adjusted lifeyears (DALYs) for 315 diseases and injuries and healthy life expectancy (HALE), 1990-2015: a systematic analysis for the global burden of disease study 2015. *The Lancet* 388:1603– 1658.
- Kotloff KL, et al. (2013) Burden and aetiology of diarrhoeal disease in infants and young children in developing countries (the global enteric multicenter study, gems): a prospective, case-control study. *The Lancet* 382(9888):209–222.
- World Health Organization (2014) Immunization summary: a statistical reference containing data through 2013, Technical report.
- Cortese MM, D. PU (2009) Prevention of rotavirus gastroenteritis among infants and children recommendation of the advisory committee on immunization practices. *MMWR* 58(RR02):1– 25.
- Dennehy PH (2013) Transmission of rotavirus and other infectious pathogens in the home. Pediatric Infectious Disease Journal 19:S103–105.
- de Blasio BF, Kasymbekova K, Flem E (2010) Dynamic model of rotavirus transmission and the impact of rotavirus vaccination in Kyrgyzstan. Vaccine 28(50):7923–32.

- Ward RL, et al. (1986) Human rotavirus studies in volunteers: determination of infectious dose and serological response to infection. *Journal of Infectious Diseases* 154(5):871–880.
- Pitzer VE, et al. (2009) Demographic variability, vaccination, and the spatiotemporal dynamics of rotavirus epidemics. *Science* 325(5938):290–294.
- Pitzer VE, et al. (2011) Influence of birth rates and transmission rates on the global seasonality of rotavirus incidence. *Journal Royal Society Interface* 8:1584–1593.
- Martinez PP, King AA, Yunus M, Faruque ASG, Pascual M (2016) Differential and enhanced response to climate forcing in diarrheal disease due to rotavirus across a megacity of the developing world. *Proceedings of the National Academy of Sciences* 113(15):4092–4097.
- Levy K, Hubbard AE, Eisenberg JNS (2009) Seasonality of rotavirus in the tropics: a systematic review and meta-analysis. *International Journal of Epidemiology* 38(6):1487–1496.
- Carlton EJ, Woster AP, DeWitt P, Goldstein RS, Levy K (2016) A systematic review and metaanalysis of ambient temperature and diarrhoeal diseases. *International Journal of Epidemiology* 45(1):117–130.
- Cook S, Glass R, LeBaron C, Ho M (1990) Global seasonality of rotavirus infections. Bulletin of the World Health Organization 68(2):1281–1289.
- Moe K, Shirley JA (1982) The effects of relative humidity and temperature on the survival of human rotavirus in faeces. Archives of Virology 72:179–186.
- Sattar SA, N LE, Springthorpe VS, Nair RC (1986) Institutional outbreaks of rotavirus diarrhoea: potential role of fomites and environmental surfaces as vehicles for virus transmission. J Hyg 96:277–289.
- Ansari S, S. SV, A. SS (1991) Survival and vehicular spread of human rotaviruses: possible relation to seasonality of outbreaks. *Rev Infec Dis* 13(3):448–461.
- Boone S, Gerba CP (2007) Significance of fomites in the spread of respiratory and enteric viral disease. Applied and Environmental Microbiology 73(6):1687–1696.
- Abad FX, Pintó RM, Bosch A (1994) Survival of enteric viruses on environmental fomites. *Applied and Environmental Microbiology* 60(10):3704–3710.
- Levy K, P WA, Goldstein RS, Carlton EJ (2016) Untangling the impacts of climate change on waterborne diseases: a systematic review of relationships between diarrheal diseases and temperature, rainfall, flooding, and drought. *Environmental Science and Technology* 50:4905–4922.
- Kolstad EHW, Johansson KA (2011) Uncertainties associated with quantifying climate change impacts on human health: a case study for diarrhea. *Environmental Health Per*spectives 119(3):299–305.
- Sharpiro RS, Cowen LE (2012) Thermal control of microbial development and virulence: mechanisms of microbial temperature sensing. *mBio* 3(5):e00238–12.
- Konkel ME, Tilly K (2000) Temperature regulated expression of bacterial virulence genes. Microbes and infection 2(2):157–66.
- Mellor JE, et al. (2016) Planning for climate change: the need for mechanisc systems-based approaches to study climate change and impacts on diarrheal diseases. Science of the Total Environment 548-549:82–90.
- Eisenberg JNS, et al. (2006) Environmental change and infectious disease: how new roads affect the transmission of diarrheal pathogens in rural Ecuador. *Proceedings of the National Academy of Sciences of the United States of America* 103(51):19460–5.
- Rao G, et al. (2015) Spatial variability of Escherichia coli in rivers of northern coastal Ecuador. Water 7(2):818–832.
- Espinosa AC, et al. (2008) Infectivity and genome persistence of rotavirus and astrovirusin groundwater and surface water. Water Research 42(10-11):2618–2628.
- Ward RL, Knowlton DR, Winston PE (1986) Mechanism of inactication of enteric viruses in fresh water. Applied and Environmental Microbiology 52(3):450–459.
- Raphael RA, A. SS, Springthorpe VS (1985) Long-term survival of human rotavirus in raw and treated river water. *Canadian Journal of Microbiology* 31(2):124–128.
- El-Sanousy WM, Osman GA, Melegy AA (2014) Survival of adnovirus, rotavirus, hepatitus A virus, pathogenic bacteria, and bacterial indicators in ground water. World Applied Sciences Journal 29(3):337–348.
- Hansen JJ, Warden PS, Margolin AB (2007) Inactication of adenovirus type 5, rotavirus WA and male specific coliphage (MS2) in biosolids by lime stabilization. *International Journal of Environmental Research and Public Health* 4(1):61–67.
- Sattar SA, Raphael RA, Springthorpe VS (1984) Rotavirus survival in conventionally treated drinking water. *Canadian Journal of Microbiology* 30(5):653–656.

- 32. Höglund C, Ashbolt N, Stenström TA, Svenson L (2002) Viral persistence in source-separated
- urine. Advances in Environmental Research 6:265–275. 33. Chung H, Sobsey MD (1993) Comparative survival of indicator viruses and enteric viruses in
- seawater and sediment. Water Science and Technology 27(3-4):425–428.
 34. McDaniels AE, Cochran KW, Gannon JJ, Williams GW (1983) Rotavirus and reovirus stability
- in microorganism-free distilled and wastewaters. *Water Research* 17(10):1349–1353. 35. Patel MM, et al. (2013) Global seasonality of rotavirus disease. *Pediatr Infect Dis J*
- 32(4):e134–e147.
 36. Rothman K, Greenland S (2005) Causation and causal inference in epidemiology. *American Journal of Public Health* 95(S1):S144–S150.
- V. A (2011) The final size of an epidemic and its relation to the basic reproduction number. Bulletin of Mathematical Biology 73:2305–2321.
- Eisenberg JN, Lewis BL, Porco TC, Hubbard AH, Colford JM (2003) Bias due to secondary transmission in estimation of attributable risk from intervention trials. *Epidemiology* 14(4):442–450.
- Glass R, et al. (2012) The control of rotavirus gastroenteritis in the United States. Transactions of the American Clinical and Climatological Association 123:36–53.
- Intergovernmental Panel on Climate Change (2001) Climate change 2001: The scientific basis. Contribution of Working Group I to the Third Assessment Report pp. 1–944.
- Deatherage DE, Kepner JL, Bennett AF, Lenski RE, Barrick JE (2017) Specificity of genome evolution in experimental populations of Escherichia coli evolved at different temperatures. *Proceedings of the National Academy of Sciences* Published online February 15, 2017:E1904– E1912.
- Bertuzzo E, et al. (2008) On the space-time evolution of a cholera epidemic. Water Resources Research 44:W01424.
- Downing JA, et al. (2006) The global abundance and size distribution of lakes, ponds, and impoundments. *Limnol Oceanogr* 51(5):2388–2397.
- 44. Central public health and environmental engineering organization (1999) Manual of water supply and treatment.
- Teunis PFM, van der Heijden OG, van der Giessen JWB, Havelaar AH (1996) The doseresponse relation in human volunteers for gastro-intestinal pathogens. *Rijksinstituut voor* volksgezondhid en milieu bilthoven 284550002:1–87.
- Brouwer A, Weir M, Eisenberg M, Meza R, Eisenberg J (2017) Dose-response relationships for environmentally mediated infectious disease transmission models. *PLoS Computational Biology* 13(4):e1005481.
- Banks R (1974) A mixing cell model for longitudinal dispersion in open channels. Water Resources Research 10(2):357–358.
- U.S. Environmental Protection Agency (2011) Exposure factors handbook 2011 edition (final), (Washington, DC, EPA/600/R-09/052F), Technical report.
- Gurwith M, W. W, Hinde D, Feltham S, Greenberg H (1981) A prospective study of rotavirus infection in infants and young children. *Journal of Infectious Diseases* 144(3):218–224.
- Molla AM, Rahman M, Sarker SA, Sack DA, Molla A (1981) Stool electrolyte content and purging rates in diarrhea caused by rotavirus, enterotoxigenic E. coli, and V. cholerae in children. *Journal of Pediatrics* 98(5):835–838.
- Fang S, et al. (2009) Dose-dependent effect of Lacobacillus rhamnosus on quantitative reduction of faecal rotavirus shedding in children. *Journal of Tropical Pediatrics* 55(5):297–301.
- World Health Organization (2006) World health organization child growth standards: length/height for age, weight-for-age, weight-for-length, weight-for-height, and body mass index-for-age. pp. 1–312.
- 53. Trummers B (2006) Datathief iii. http://datathief.org/.
- Crane SR, Moore JA (1986) Modeling enteric bacterial die off: a review. Water, air, and soil pollution 27:411–439.
- Brouwer AF, et al. (2017) Modeling biphasic environmental decay of pathogens and implications for risk analysis. *Environmental Science and Technology* 51(4):2186–2196.
- Van den Driessche P, Watmough J (2002) Reproduction numbers and sub-threshold endemic equilibria for compartmental models of disease transmission. *Mathematical biosciences* 180(1):29–48.
- Gatto M, et al. (2013) Spatially explicit conditions for waterborne pathogen invasion. The American Naturalist 182(3):328–346.